Author's response to reviews

Title: Routine care provided by specialists to children and adolescents in the United States: 2002-2006

Authors:

Jose M Valderas (jose.valderas@manchester.ac.uk)
Barbara Starfield (bstarfie@jhsphs.edu)
Christopher B Forrest (FORRESTC@email.chop.edu)
Luis Rajmil (lrajmil@imim.es)
Martin Roland (martin.roland@medschl.cam.ac.uk)
Bonnie Sibbald (bonnie.sibbald@manchester.ac.uk)

Version: 2 Date: 16 June 2009

Author's response to reviews:

No revisions included, this is the first submission.